\begin{table}%T1
\caption{Journal of observations and parameters of the studied stars.}
\label{tab:obs}
%\centerline
{\small\begin{tabular}{@{}@{\extracolsep{\fill}}rllccrrccrlr
@{\extracolsep{0pt}}@{}}
\hline\hline\\[-6pt]
&&&&&&&Fitted&Litt.\\
\multicolumn{1}{c}{HD}&Other&Spectral&HJD&$S/N$
&\multicolumn{1}{c}{\Teff}&Ref.&\vsi&\vsi&Ref.&Rotation&Ref.\\
&identification&type&$\null-2~400~000.$&
&\multicolumn{1}{c}{(K)}&&(km~s$^{-1}$)&(km~s$^{-1}$)&&period\\[4pt]
\hline\\[-4pt]
 47103&BD +20\degr1508   &Ap SrEu   &50~115.615&160& 8900& 1&$4.0\pm0.3$&$4.4\pm0.5$ & 1&         &  \\
 81009&HR 3724           &A3p CrSrSi&50~115.681&245& 8500& 2&$5.0\pm0.5$&$\null\leq$5& 2&33\fd984 & 3\\
 91375&HR 4138           &A2IV      &50~243.473&285& 9300& 4&$7.0\pm0.5$&6.5         & 4&         &  \\
 94660&HR 4263           &A0p EuCrSi&50~115.721&340&10~800& 5&$7.0\pm1.0$&            &  &2800~d   & 3\\
 96446&CPD $-$59\degr3038&B2 He str &50~115.740&360&22~000& 6&           &$0\pm4$     & 6&0\fd85137& 7\\
116114&BD $-$17\degr3829 &F0p SrCrEu&50~115.774&180& 7300& 5&$4.5\pm0.5$&            &  &27\fd61  & 3\\
116458&HR 5049           &A0p SiEuCr&50~115.790&260&10~300& 5&$1.0\pm1.0$&4           & 8&148\fd39 & 3\\
133792&HR 5623           &A0p SrCr  &50~244.476&410& 8800&  &$1.0\pm0.5$&            &  &         &  \\
137949&33 Lib            &F0p SrEuCr&50~115.806&230& 7350& 9&$4.0\pm0.5$&            &  &$\null\ga$75~y?&10\\[3pt]
\hline\\[-4pt]
\end{tabular}}
{\em References:\/}
(1)~Babel et~al. (\cite{BNQ95}); (2)~Wade et~al. (\cite{W00});
   (3)~Mathys et~al. (in preparation); (4)~Lemke (\cite{L89});
(5)~Hubrig et~al. (\cite{HNM00}); (6)~Zboril \& North (\cite{ZN99}); 
(7)~Matthews \& Bohlender (\cite{MB91}); (8)~Nielsen \& Wahlgren
(\cite{NW00}); (9)~Ryabchikova et~al. (\cite{R99}); (10)~Mathys et~al. (\cite{MHLLM97}). 
\end{table}